November 17, 2011

The BMC Series Editorial Production Team
BioMed Central
U.K.

RE: Manuscript No. 9749544196081329
Role of Aldosterone on Lung Structural Remodelling and Right Ventricular Function in Congestive Heart Failure

Please find enclosed the proof version of the above-named manuscript. As suggested by the Editorial Production Team, we have deleted the text "short title" from the title page. The list of authors in the manuscript is written exactly as they are in the submission system, both in style and order. We have also indicated in the "Authors' contributions" section that all authors have read and approved the final manuscript.

Sincerely,

[Signature]

JD:lb

Jocelyn Dupuis, MD, PhD
Email: jocelyn.dupuis@bellnet.ca

encl.